Author's response to reviews

Title: Treatment needs and acknowledgement of illness. Importance for satisfaction with psychiatric inpatient treatment

Authors:

Knut W. Soergaard (ksorgaar@online.no)
Mary Nivison (mary.nivison@unn.no)
Vidje Hansen (vidje.hansen@unn.no)
Terje Oeisvold (terje.oeiesvold@nilsh.no)

Version: 4 Date: 5 May 2008

Author's response to reviews:

Email and affiliations: The email addresses and the affiliations are separated, email addresses are preceded by the authors initials.

Figure cropping: completed

05 05 2008